Author's response to reviews

Title: Distribution and transmission of Mycobacterium tuberculosis complex lineages among children in peri-urban Kampala, Uganda

Authors:

Eddie M Wampande (wamps@covab.mak.ac.ug)
Ezekiel Mupere (mupez@yahoo.com)
Devan Jaganath (djaganath@mednet.ucla.edu)
Mary Nsereko (mnsereko@mucwru.or.ug)
Harriet K Mayanja (hmk@chs.mak.ac.ug)
Kathleen Eisenach (eisenachkathleen@uams.edu)
Henry W Boom (whb@case.edu)
Sebastien Gagneux (sebastien.gagneux@unibas.ch)
Moses L Joloba (mlj10@case.edu)

Version: 6 Date: 24 August 2015

Author's response to reviews: see over
To the Journal Editorial office
BMC Pediatrics

Dear Sir/Madam
We write to thank you for accepting our manuscript titled: **Distribution and transmission of Mycobacterium tuberculosis complex lineages among children in peri-urban Kampala, Uganda.** In this paper we report that *Mycobacterium tuberculosis* lineages circulating in peri-urban Kampala Uganda transmit equally. Furthermore, we also thank you for the reviewers’ comments that we received. Fortunately, both reviewers (reviewer 1: Dr. Ehsan Aryan and reviewer 2: Dr. Laura Rindi) indicated that the manuscript doesn’t require any revision and it is suitable for publication in its current form.

Thank you,

Eddie Wampande

*On behalf of the co-authors*